Author's response to reviews

Title: Fatigue in osteoarthritis: a qualitative study

Authors:

J. Denise Power (dpower@uhnres.utoronto.ca)
Elizabeth M Badley (e.badley@utoronto.ca)
Melissa R French (melissa.french@wchospital.ca)
Angela J Wall (angela.wall@wchospital.ca)
Gillian A Hawker (gillian.hawker@wchospital.ca)

Version: 5 Date: 23 April 2008

Author's response to reviews:

Dear Editor,

The following typographical errors have been corrected in our attached revised manuscript.

Page 1 - Author initials for Gillian A Hawker have been corrected to GAH

Page 19, Reference 6. Duplication of the word "the" before World Bank has been deleted

Page 19, Reference 7. The date of publication (2003) has been added to this citation

Page 20, Reference 14. Full citation details [45(7) 885-889] have been added

Page 21, Reference 29. Spelling of author's name, Buchanan, has been corrected.

No further changes are requested.

Thank you.